%&LaTeX
\documentclass{article}

\usepackage[utf8]{inputenc}
\usepackage[T1]{fontenc}
\usepackage{textcomp}

\begin{document}

\section*{2005}

Misael Rosales, Petia Radeva, Oriol Rodriguez, \& Debora Gil. (2005).<a target={\textquoteright}_blank{\textquoteright} href={\textquoteright}/files/RRR2005.pdf{\textquoteright}> "Suppression of IVUS Image Rotation. A Kinematic Approach "</a> In Monica Andres and Hernandez Petia and Santos A. and R. Frangi (Ed.), \textit{Functional Imaging and Modeling of the Heart} (Vol. 3504, pp. 889--892). Lecture Notes in Computer Science, 3504. Springer Berlin / Heidelberg.

\end{document}

